This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath

This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath This is a sample of TEX output using nmath